Author's response to reviews

Title: On the pedagogy of pharmacological communication: A study of final semester health science students

Authors:

   Ann Zetterqvist (ann.zetterqvist@ped.gu.se)
   Patrik Aronsson (patrik.aronsson@pharm.gu.se)
   Staffan Hägg (staffan.hagg@liu.se)
   Karin Kjellgren (karin.kjellgren@liu.se)
   Margareta Reis (margareta.reis@liu.se)
   Gunnar Tobin (gunnar.tobin@pharm.gu.se)
   Shirley Booth (shirley.booth@gu.se)

Version: 4 Date: 12 October 2015

Author's response to reviews:

Dear Editor!

We have now revised our manuscript based on your last three comments:

1) The full name and affiliation of the ethics committee that approved the study:
   An ethical approval is not required in Sweden for this type of study according to the scientific secretary of the Regional Ethical Review board in Linköping.

2) State in the Methods section whether written informed consent for participation in the study was obtained from participants or, where participants are children, a parent or guardians and whether it was verbal or written:
   The paragraph “The study does not involve any handling of sensitive personal data or clinical procedures. All interviewees gave their informed consent prior to their inclusion in the study. The project complies fully with current applicable Swedish legal rules and ethical guidelines including the Helsinki declaration.” has been inserted in the Methods section.

3) Include a heading for "Authors' Contributions":
   Done.

Best wishes
Ann Zetterqvist (corresponding author)